Author's response to reviews

Title: Melatonin analgesia is associated with improvement of the descending endogenous pain-modulating system in fibromyalgia: A phase II, randomized, double-dummy, controlled trial

Authors:

Simone A Zanette (simonezanette@yahoo.com)
Rafael Vercelino (ryvercelino@gmail.com)
Gabriela Laste (gabrielalaste@gmail.com)
Joanna R Rozisky (joannarozisky@gmail.com)
André Schwertner (andreschwertner@yahoo.com.br)
Caroline B Machado (simonezanette@yahoo.com)
Fernando Xavier (fernando.coxavier@gmail.com)
Izabel CS de Souza (belcustodio20@yahoo.com.br)
Alicia Deitos (aliadcitos@gmail.com)
Iraci LS Torres (iracitorres@gmail.com)
Wolnei Caumo (caumo@cpovo.net)

Version: 2
Date: 17 March 2014

Author's response to reviews:

New version was included:

1. Title Page, with the email addresses of all authors.
2. Line and page numbers.
3. Conclusions after Discussion.